Author's response to reviews

Title: Gender differences in self reported long term outcomes following moderate to severe traumatic brain injury

Authors:

   Angela Colantonio (angela.colantonio@utoronto.ca)
   Jocelyn E Harris (jocelyn_ellen@hotmail.com)
   Graham Ratcliff (graham_ratcliff@hotmail.com)
   Sue Chase (chase.susan@gmail.com)
   Kristina Ellis (ellis.kristina@torontorehab.on.ca)

Version: 4 Date: 26 October 2010

Author's response to reviews:

Editoral Formatting changes to be made:

Major revisions:

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript. DONE

Tables- please remove the visible vertical lines from your tables. DONE

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Introduction: Please rename this section 'Background'. DONE